Title: The prognostic value of combined TGF-beta1 and ELF in hepatocellular carcinoma

Authors:

Fei Ji (jifei1215@163.com)
Shun-Jun Fu (fsj103@163.com)
Shun-Li Shen (shunlishen@163.com)
Long-Juan Zhang (zhlongji@mail.sysu.edu.cn)
Qing-Hua Cao (bpigeon@163.com)
Shao-Qiang Li (Lisg@medmail.com.cn)
Bao-Gang Peng (pengbaogang@medmail.com.cn)
Li-Jian Liang (dr_liangli@163.com)
Yun-Peng Hua (hyp0427@163.com)

Version: 4
Date: 10 February 2015

Author's response to reviews: see over
Dear Editor:

We are pleased to resubmit our manuscript entitled “The prognostic value of combined TGF-β1 and ELF in hepatocellular carcinoma” for your consideration for publication as original article in *BMC Cancer* again.

We revise our manuscript to highlight all the changes made in the revised manuscript, all the changes have been marked with red color in the manuscript. After revision, we submit our files with the changes requested again.

Thank you for your time and kind consideration, Thank you very much.

Best regards

Yours Sincerely

Prof. Yunpeng Hua

Department of liver Surgery
The First Affiliated Hospital of Sun Yat-Sen University
58, Zhongshan Road 2, Guangzhou 510080, China.
Email: hyp0427@163.com
Telephone: 86-20-87755766 ext 8096.
Fax: 86-20-87755766 ext 8663